Author’s response to reviews

Title: Perceived organizational support and moral distress among nurses

Authors:

Navideh Robaee (Navideh16@gmail.com)
Foroozan Atashzadeh –Shoorideh (f_atashzadeh@sbmu.ac.ir)
Tahereh Ashktorab (t.ashktorab@sbmu.ac.ir)
Ahmadreza Baghestani (baghestani.ar@gmail.com)
Maasoumeh Barkhordari-Sharifabad (barkhordary.m@gmail.com)

Version: 3 Date: 06 Dec 2017

Author’s response to reviews:

Dear Editor of Journal of BMC Nursing

Thank you for the opportunity to revise our manuscript.

1. In your “ethical approval and consent to participate” section please confirm whether written consent was obtained from participants. If written consent was not obtained, please explain why not.

It was added (Page 16, line 7).

2. Please include a title page as the first page of your manuscript where you list the full names, institutional addresses and email addresses for all authors. The corresponding author should be indicated.

It was done.

3. Please remove your response to the reviewers comments from your file inventory as it is no longer needed

It was removed.
4. We would also like to ask for you to provide more justification for the contributions of MB and AB as currently they do not automatically qualify for authorship. Contribution to the listed tasks alone, does not usually justify authorship.

It was added (the first line in Authors’ contributions section)

5. Please remove the funding information from the Acknowledgements and include it in the Funding section instead. If you have no further acknowledgements please put “Not Applicable” in the Acknowledgements section.

It was removed.

6. In the Funding section, please also describe the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

It was added.

7. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

It was done.

Regards,

Dr. Foroozan Atashzadeh –Shoorideh, Associate Professor, Department of Nursing Management, Nursing & Midwifery School, Shahid Beheshti University of Medical Sciences, Tehran, Iran. E-mail: f_atashzadeh@sbmu.ac.ir